# 21 WORKING PAPERS 2024

## EXPORT PROMOTION WITH MATCHMAKING AND GRANTS: EVIDENCE FROM PORTUGUESE FIRMS

Paulo Barbosa | João Amador Esmeralda Arranhado



# 21 WORKING PAPERS 2024

## EXPORT PROMOTION WITH MATCHMAKING AND GRANTS: EVIDENCE FROM PORTUGUESE FIRMS

Paulo Barbosa | João Amador Esmeralda Arranhado

DECEMBER 2024 The analyses, opinions and findings of this paper represent the views of the authors, and not necessarily those of the Banco de Portugal or the Eurosystem.

> Please address correspondence to Banco de Portugal Rua do Comércio 148, 1100-150 Lisboa, Portugal Tel.: +351 213 130 000, email: info@bportugal.pt



Lisboa, 2024 • www.bportugal.pt

Working Papers | Lisboa 2024 • Banco de Portugal Rua do Comércio 148 | 1100-150 Lisboa • www.bportugal.pt • Edition Banco de Portugal • ISBN (online) 978-989-678-915-2 • ISSN (online) 2182-0422

#### Export Promotion with Matchmaking and Grants: Evidence from Portuguese Firms

Paulo Barbosa ISEG - University of Lisbon **João Amador** Banco de Portugal Nova SBE

**Esmeralda Arranhado** ISEG - University of Lisbon

11 December 2024

#### Abstract

This paper estimates the impact on firms' goods exports of two key policies implemented by export promotion agencies (EPAs): matchmaking in international markets and financial grants for internationalization. We merge Portuguese rich firm-level data on exports of goods, balance sheets, and income statements, with detailed information on the activity of the EPA between 2012 and 2021. The empirical exercise estimates the causal effect of these policies with a staggered difference-in-difference estimator. We conclude that the support provided to Portuguese firms significantly and positively affected their exports of goods. Financial grants for internationalization led to a significant increase in firms' goods exports, with a greater effect on micro and small firms and in the sectors of "Wholesale of household goods" and "Manufacture of wearing apparel". Similarly, matchmaking activities led to a significant increase in exports. In France, the country for which Portuguese companies request the most matching support, the effect is greater for micro and small firms and in the sector of "Manufacturing of other textiles".

JEL: D22, F13, F14, L25, L53 Keywords: Export promotion, Staggered difference-in-differences, Firm-level data, Portugal.

Acknowledgements: The opinions expressed in this paper are those of the authors and do not necessarily coincide with those of Banco de Portugal or the Eurosystem. E. Arranhado was partially supported by the Project CEMAPRE/REM - UIDB/05069/2020 - financed by FCT/MCTES through national funds. Authors thank participants in ISEG's seminar for very useful comments and suggestions. Any remaining errors and omissions are the sole responsibility of the authors. Address: Banco de Portugal, Economics and Research Department, R. Francisco Ribeiro 2, 1150-165 Lisboa, Portugal.

E-mail: paulofbarbosa5@gmail.com; jamador@bportugal.pt; earranhado@iseg.ulisboa.pt

#### 1. Introduction

Venturing into a new foreign market is a very significant decision for any firm. There are potentially high entry costs associated with establishing a commercial presence abroad, and the outcome of this move is quite uncertain. Although it is up to firms to perform such cost-benefit analysis and decide, Export Promotion Agencies (EPAs) may significantly reduce sunk costs and uncertainties by providing financing or information on potential markets. A positive outcome from EPAs actions would be a large number of successful entries into foreign markets, thus increasing exports and, ultimately, trade gains, employment, and economic growth.

The most prevalent strand of trade models considers heterogeneous firms that choose to be exporters if their productivity is high enough to cover all costs involved in the process (Roberts and Tybout 1997; Melitz 2003; Melitz and Redding 2014). This approach suggests a strong self-selection of the most efficient firms into being exporters, meaning that they would be larger and pay higher wages even without being present in the export market. Nevertheless, several papers have shown that market size, i.e., firms' broader access to foreign markets, strongly influences their innovation and productivity. Therefore, governments' efforts to improve firms' access to foreign markets could lead to more investment in innovation and higher productivity (De Loecker 2007; Harrison and Rodríguez-Clare 2010). This is referred to in the literature as the learning-by-exporting argument. In this vein, Lileeva and Trefler (2010) examined the effect of the cut in US tariffs on Canadian products and showed that firms that started exporting or that exported more after this event: (i) increased labor productivity; (ii) engaged in more product innovation; and (iii) adopted more advanced manufacturing technologies. Atkin et al. (2017) also report empirical evidence of large gains from *learning-by-exporting*. The paper selected a random sample of handmade carpets producers from a region of Egypt to whom was provided the opportunity to export to high-income markets. Authors reported that treated firms decreased output, but increased prices, total hours worked, and profits. The profit increase was an expected outcome because there is a demand shock for the treated firms. Nevertheless, authors argue that the increase in profits due to the supply-side effect was higher than what is typically found in the literature.<sup>1</sup> These findings suggest that high-income countries demand high-quality products that take longer to produce, thus supporting the *learning-by-exporting* narrative.

Increasing the number of firms selling abroad has been a key objective of policymakers. During the last two decades, EPAs became the most popular institutions helping firms to turn into successful exporters (Srhoj *et al.* 2023). EPAs are mostly public institutions with a significant budget, but there are private firms providing similar services. In this context, an important strand of research is

<sup>1.</sup> See Banerjee (2013) for an example of supply side effects arising from different access to credit.

the cost-benefit analysis of such government policies, including the impact of EPAs on firms' productivity (see Cruz *et al.* (2018) and Srhoj *et al.* (2023) for a survey).

EPAs support firms to internationalize by helping them to overcome export barriers. The economic argument for government funding of EPAs is the existence of market failures, such as incomplete information, and positive externalities associated to exporting firms. Firstly, learning about foreign markets is a way of reducing the underlying uncertainty associated to starting to export. The fixed and/or sunk costs of acquiring such information may be large for an individual firm, thus discouraging entry in a foreign market. EPAs face lower average costs in this process because information about foreign markets can be shared across many firms. Secondly, the positive externalities emerging from exporters relate to spillover effects on other firms with similar products. These effects can arise from reputation for product quality and linkages in demand. For a comprehensive discussion of the arguments that support export promotion policies see Copeland (2007) and Aalto and Gustafsson (2020).

Broocks and Van Biesebroeck (2017) and Munch and Schaur (2018) investigate the effect of matching services provided by EPAs on the exports of firms in Flanders (the largest Belgian region) and Denmark, respectively. The former paper shows that firms benefiting from such support had an 8.5 percentage point (pp) higher probability of exporting to an extra-EU market than those without any EPA's support. The latter paper shows that for small firms, the matchmaking service of the Danish EPA increased the probability of entry in foreign markets. Sales and value-added increased by 8 pp, while employment increased by 4 pp relative to control firms. Moreover, the effects of the EPAs' support were higher for micro, small and medium firms.

The literature about the effect of EPAs on firms' performance is scarce. The main reason is the very strong data requirements for an econometric exercise. It is typically necessary to merge three types of data: (i) EPA data on services provided to firms; (ii) firm-level trade transaction data by destination and product; and (iii) control variables, commonly reported in balance sheets and income statements. We were able to merge this type of firm level data for Portugal for the period 2012-2021 using the firm unique identifier. This unique set of data derives from AICEP's firm-level information about time and type of support provided to firms, plus firm-level data on trade transactions by *Statistics Portugal*, the national statistics agency, and the Simplified Business Information database, the official business register, containing information on firm's characteristics.

Our paper contributes to the literature on the evaluation of the impact of EPAs on firms' performance, specifically on their exports. We use an event study methodology to assess the effect of the two main services provided by the Portuguese EPA – Portugal Trade & Investment (AICEP in the Portuguese acronym) – (i) grants for internationalization, and (ii) matchmaking in international markets. The financial aid for firms' internationalization is generally attributed to cover expenses such as trade missions, attending international fairs, advertisement in international markets, and support production upscale. The matchmaking service

consists of providing detailed information about potential importers of Portuguese products. AICEP's offices worldwide collect detailed information about firms that may be prone to import Portuguese products and provide this information in tailored reports. In order to have a sufficient number of observations we measure the matchmaking effect on exports to the country for which there is the highest demand for this service – France. Although it can be argued that matchmaking services for a country can trigger exports elsewhere, the first order effect should occur on the country surveyed. As for grants, which may be used for different purposes, we assess the total impact on firms' total exports.

The methodological challenge is to prove causality from export promotion policies to firm's export performance. Indeed, it is possible that firms that are already good performers in international markets apply for grants and require matchmaking services. These firms may be the ones most aware about the potential benefits of export promotion policies because they have used them earlier. In addition, access may involve costs or the prior fulfillment of requirements, which bias certain types of firms to apply for these funds and services. Indeed, Portuguese firms that benefit from grants are those with relatively higher exports, while those requesting matchmaking services export less than the ones not using this service.

Our database covers the period from 2012 to 2021 and the treatment associated with grants and matchmaking services is staggered, i.e., different firms benefit from it in different moments. Moreover, treatment is a binary variable as no information on the amount of grants or intensity of the matchmaking service is available. The impacts of the treatment are obtained using the staggered Differences-in-Differences (DiD) interaction-weighted estimator proposed by Sun and Abraham (2021).<sup>2</sup>

Usually, matchmaking and, especially, grants are the highest level of support provided by EPAs. For this reason, access to these services requires some degree of conditionality. Thus, to optimize resource utilization, EPAs typically support firms that have already shown basic competence to perform in foreign markets. Virtually almost all firms in the AICEP database record positive exports, and those requesting matchmaking services to France are, in general, already exporting. Therefore, our analysis only refers to the impact of promotion on the intensive margin of exports. Although the extensive margin of exports in the dimensions of the firm, destination, and product are relevant in terms of policy action, other tools are used for this purpose, and this margin does not explain much of the yearly export developments (Amador and Opromolla 2013, 2017).

Our results are robust and suggest that effects of export promotion policies are sizeable. Grants led to an overall increase of about 13% in firms' total exports. The effect is larger for micro and small firms, where financial aid increased exports by 18.9%. In sectors "Wholesale of household goods" and "Manufacture of wearing

<sup>2.</sup> For recent contributions on the applicability of staggered DiD estimators see Baker *et al.* (2022) and de Chaisemartin and D'Haultfœuille (2020).

apparel" the effect is sizeable, about 36% and 47%, respectively. The analysis of AICEP's matchmaking service for France suggests an increase of about 32% in firm's exports to this country. Furthermore, there is a significant size and sector heterogeneity. The effect on micro and small firms is about 40%, and for firms in the "Manufacture of other textiles", matchmaking led to an increase of 137% in firm's exports.

The paper is organized as follows. Section 2 overviews the relevant literature, describes the Portuguese EPA and details the two export promotion policies under analysis. Section 3 presents the data sources and characterizes firms taking up grants and asking for matchmaking services to enter or increase sales in foreign markets. Section 4 describes the identification strategy used and robustness tests. Results are presented in Section 5, which is divided in two. Subsection 5.1 presents the estimates of the effect of the financial support and the matchmaking service and Subsection 5.2 details the results by sector and size. Section 6 presents the results of the robustness tests. Finally, Section 7 presents some concluding remarks.

#### 2. Background on export promotion

#### 2.1. Literature review

The effectiveness of export promotion policies (EPP) has been studied along two main strands of research: (i) at the aggregate level, evaluating the effect of trade missions and the effect of diplomatic offices on bilateral trade flows, and (ii) using firm-level data, with information about the EPAs' support and firms' characteristics as controls.<sup>3</sup> The first strand of research reports ambiguous effects. Bernard and Jensen (2004) use a panel of US manufacturing plants and find no link between government expenditure on export promotion and the probability of entry into exporting. Head and Ries (2010) analyzed the trade missions carried out by the Canadian government and found no evidence that they increased trade with the respective country. Other aggregate level studies find a significant effect of EPP on exports. Rose (2007) reports a positive effect of opening an embassy on export performance in the US. Similarly, Gil-Pareja et al. (2015) shows a positive effect from regional export promotion offices on aggregate exports in Spain. For Japan and Korea Hayakawa et al. (2014) also found a positive effect of opening an EPA. The paper concludes that opening an EPA in a country has a similar effect to establishing a free trade agreement with the same country.

Firm-level studies on the impact of grants for entering international markets report a positive effect. Comi and Resmini (2020) take a DiD approach to investigate the impact of several EPPs – grants, vouchers for attending

<sup>3.</sup> Van Biesebroeck *et al.* (2016) and Srhoj *et al.* (2023) offer a theoretical discussion and a summary of empirical results on the effectiveness of EPPs.

international fairs, trade missions, and external counseling – in Italy. They show that firms assisted by these EPPs presented higher export propensity and export intensity when compared with non-assisted firms, with higher effects on micro and small firms, as well as on those already exporting. Srhoj and Walde (2020) uses Croatian firm-level data to analyze the effect of an export-oriented grant, originally designed to facilitate the accession to the European Union. These grants aimed to support firms in two main activities: (i) commercial activities, for example contract consulting services for entry in international markets and attending international fairs, and (ii) technological development. The paper finds that the EPPs positively affected firms' export value, total sales, value-added, and profit, but not employment. Grants for technological development had a higher impact than commercial activities. The cost-benefit analysis indicates that the value added created by the EPPs is 39.5% larger than costs involved.

Cruz (2014) studied if firms supported by the Brazilian EPA (*Apex*) have a higher chance of entering an international market. Using a matching DiD and fixed effect estimation procedures, the author estimates that the support led to a market entry probability around 2.5 times higher for treated firms. Similarly, Makioka (2021) investigated the causal effect of a matching service on exports of Japanese firms. More specifically, the author studied the effect of attending a trade fair within an export support program provided by the Japanese EPA. The author concludes that the support provided increased the export probability by 9.6 pp in the year following the fair and exports increased by 25.4%.

In a different vein, Van Biesebroeck *et al.* (2016) investigate whether EPAsupported firms during the global recession of 2009 performed better and recovered faster. The paper uses matching of propensity scores on firms supported by the Flanders Investment and Trade (*FIT*) and Peru's National Public Export Promotion Organization (*PROMPERU*) from 2006 to 2011. The services provided by these EPAs were the resolution of local specific transaction problems, subsidies for foreign market prospecting, market information, and help in attending international fairs. Firms that received EPAs' help performed better during the crisis, i.e., kept higher export levels, were more likely to survive on foreign markets, and recovered faster after the crises. For Belgium, the effect was larger for exports outside the EU. Furthermore, the cost-benefit analysis showed that these EPPs covered their costs.

Van Biesebroeck *et al.* (2015) performs an empirical exercise similar to ours. They evaluated the trade promotion services provided by the Canadian Trade Commissioner Service (*TCS*), namely market prospect information, key contacts search, local company information, visits information, face-to-face briefing and trouble shooting. They merged three datasets containing all active exporters between 1999 and 2006, information on firms' exports, business register, and the TCS's clients dataset. Authors use a DiD estimator and find that these services increased exports to countries already served by the firms by 17.9% on average. However, they did not find an effect on the expansion of new products or the increase of new destinations for exports. Furthermore, the EPP takes a few years to

significantly affect the intensive margin of exports, and this effect is more significant in older firms and in those that export to a larger number of markets.

Similarly, Buus *et al.* (2022) studies the effect of matchmaking services provided by the Danish EPA. Although they report that the matchmaking increased firms' exports within markets, the service did not affect prices, quality, markups, or marginal costs. The authors also provide a detailed discussion on how EPAs impact firms' employment, sales, and productivity.

#### 2.2. The Portuguese Export Promotion Agency - AICEP

AICEP originates from the Portuguese Agency for Investment (*API*) created in 2002. *API* mainly targeted at the reduction of framework costs, thus enabling larger FDI flows. In 2007, *API* was renamed AICEP, also absorbing the functions of a public institute for investment and foreign trade (*ICEP*). This institute had been created in 1982 and focused on export promotion. The role of AICEP is similar to other EPAs (Cruz *et al.* 2018; Srhoj *et al.* 2023). It is a government agency funded by public resources with offices in close to 50 countries. By 2019, the agency had 464 employees, of which 318 were working in Portugal and 138 abroad. Out of these 138, 11 were in Africa, 10 in North America, 10 in South America, 17 in Asia and Oceania, 67 in Western Europe, 16 in Eastern and Central Europe, and 7 in the Maghreb and Middle East (Castela 2021).

AICEP offers a wide range of services to Portuguese firms, aiming to increase their presence in foreign markets. The set of policies includes grants for entry and expansion in foreign markets, publication of market studies and surveys, training, information on technical regulations, financing firm's presence in trade fairs, exhibitions and missions, and matchmaking in foreign markets, notably through meetings, maintenance of a detailed contact database, and other assistance. AICEP is both an EPA and an IPA, thus it also promotes Portugal as an attractive investment destination, carrying out activities generally categorized in four categories: image building, investment generation, investment facilitation and retention, and policy advocacy. Some examples of these activities are marketing on social media and international fairs, investor targeting, administrative support, and facilities to enable the dialogue between government and investors, either to solve problems and influence policies<sup>4</sup>.

As previously referred, our research delves into the pivotal role of AICEP's services for Portuguese firms that aim to increase exports, namely grants for internationalization and matchmaking in specific markets. Indeed, these two types of support are the ones most often requested by Portuguese firms, and we briefly describe them in the next two subsections.

<sup>4.</sup> For a detailed discussion on the role of IPAs, see OCDE (2018)

#### 2.3. Grants for internationalization

AICEP supports companies that aim to enter foreign markets or expand existing exports. The financial aid of AICEP is earmarked for the internationalization of firms, to facilitate their participation in international fairs and trade missions, to finance marketing campaigns abroad, and increase production. These grants are financed by the European Regional Development Fund (ERDF) and are allocated by AICEP to the most promising internationalization projects presented by the firms.

The total amount of grants attributed in the average of the period 2012-2021 was 63 million euros, and the average amount of each grant was 200 thousand euros. Grants are attributed in connection with an investment proposals by firms and its amount corresponds to about 40% of the investment amount.

In terms of information available, we know the treatment status of each firms in each year. Some firms in the sample received this financial support more than once during the period analyzed (2012-2021). We do not know to which country (countries) firms directed the funds, that is, in which country the firms spent the grant for market entry or expanding its sales. Thus we evaluate the effect of grants on firms' total exports.

#### 2.4. Matchmaking in export markets

AICEP's matchmaking service operates bidirectionally: firms can request matchmaking through their key account manager (KAM), who can also proactively offer this service. We do not have information on whether a firm requested the service or if it was offered by the KAM. After the firm requests the service or it is offered by its KAM, AICEP's headquarters in Portugal contacts its respective office abroad, which then compiles the information into a customized report that is delivered to the firm. The report includes details on potential importers of Portuguese products. Each firm can request this service for up to three markets at once. For example, a company considering the expansion of sales in Germany, France, Brazil, and China must select three countries, and request the service for the fourth country only when it receives one of the three already ordered. On average, 20 business days elapse from the moment the service is requested until the report is received.

Our evaluation of AICEP's matchmaking services relies on the list of markets for which individual firms asked information about, together with data on firmlevel exports to each market and additional control variables. The impact of the matchmaking service is assessed as the direct impact on the value of firm's exports to the market surveyed.

Our analysis focuses on the French market, which posts the largest number of observations. In the period 2012-2021 the number of matchmaking requests for France reached 744, with an average of 60 firms involved per year. Germany, UK,

USA and Spain come next in the ranking, but at some distance, with a total of 617, 592, 558 and 503 requests in the period 2012-2021, respectively.

#### 3. Data

#### 3.1. Data sources

The paper is based on a rich and unique firm-level dataset obtained from merging information from three sources: (i) the dataset on exports of firms; (ii) the balance sheet and income statement information; and (iii) the information of the AICEP client dataset.

Statistics Portugal (*Instituto Nacional de Estatística*) compiles the export database, including detailed information on export and import transactions of goods by firms located in Portugal, which is used to compute the official international trade statistics. The data originally comes from customs returns forms for extra-EU trade, and for intra-EU trade, it comes from the Intrastat reports. The dataset includes all transactions with non-EU countries. However, the intra-EU transactions are subject to the assimilation threshold, i.e., the annual trading value below which Intrastat declaration is not mandatory. The assimilation thresholds in Portugal are 350 thousand euros for EU imports and 250 thousand euros for EU exports.

The Simplified Corporate Information (Informação Empresarial Simplificada -IES) is maintained by Statistics Portugal. The IES data set contains the annual mandatory information that firms must report to tax administration, social security and statistical authorities. The information consists of economic, financial, and accounting balances for the respective fiscal year and covers the population of Portuguese non-financial corporations. Firms report detailed balance sheet items and information concerning other key variables such as the number of employees, cost of inputs and turnover.

Finally, the AICEP dataset includes information about its clients and the services provided to them. The dataset contains detailed information about the timing of the services and, for the case of grants, the country to which the firms requested support, as discussed in Section 2.2.

#### 3.2. Treatment condition and control firms

To mitigate the heterogeneity caused by varying intensities of treatment among firms, we restricted our analysis to firms that received AICEP support for a maximum of two consecutive years. For example, a firm receiving AICEP support in both 2015 and 2016 is included in our treatment group. However, a firm receiving support in 2015 and again in 2017 is excluded from our sample. This approach ensures that firms with different levels of treatment exposure, such as those receiving support intermittently over several years, do not bias results. We consider the support provided within two consecutive years as a single treatment.

This is reasonable, since a firm can receive the matchmaking and only use it within a few months. In addition, the methodological approach followed assumes that the treatments are absorbing, i.e., once the treatment occurs, the firm is classified as treated afterwards. This corresponds to a situation where the impact of the intervention at a given moment prevails in time (Sun and Abraham 2021).

AICEP offers matchmaking services to its clients and does not set strict criteria to offer this support. Nevertheless, this service is usually provided to firms with a higher potential to increase sales in international markets, and that have the capacity to pay the inherent sunk costs. In contrast, grants for internationalization are available through a public call open to all Portuguese firms. Due to the limited resources, AICEP allocates grants to the most promising internationalization projects, specifically those deemed by AICEP analysts to have the greatest potential to increase exports.

To assess the effect of matchmaking and grants, we take Portuguese goods exporters that have not received AICEP support as the control group. This approach is correct for grants because they are available to all Portuguese firms, not just AICEP clients. The control group is also suitable for matchmaking services because AICEP served approximately 65% of Portuguese goods exporters in the period, meaning that these firms could have requested matchmaking support anytime.

#### 3.3. Descriptive statistics

Table 1 presents the average and median export values for non-treated firms and firms that benefited from financial support (columns 1 to 3) and matchmaking services (columns 4 to 6), along with their respective numbers, for the sample period 2012-2021. The proportion of treated firms is small for both policies. In 2012, a mere 0.52% of firms in the dataset received grants, while only 0.19% received matchmaking support in France. The small proportion of treated firms persists throughout the analysis period. The treatment is cumulative, meaning that a firm treated in 2012 remains classified as treated in subsequent years. Considering this cumulative effect, by 2021, the percentage of firms benefiting from grants increased to 6.75%, while those receiving matchmaking services rose to 11.45%.

Additionally, Figures A.1 and A.2 in the Appendix illustrate the yearly mean of logged exports during the sample period and the Kernel distribution of export values for both treated and untreated firms. These plots reveal distinct profiles for the firms benefiting from the two services provided by AICEP. Firms with higher export volumes received grants for internationalization, whereas those with lower export volumes benefited more from matchmaking services in France. However, these differences diminished from 2012 to 2021. In 2012, compared to untreated firms, the average export value of firms receiving grants was 64.6% higher, while for those receiving matchmaking services, it was 59.1% lower. By 2021, these percentages had changed to 57.4% higher and 79.4% higher, respectively, with the matchmaking service showing a median export value higher for treated firms than for control firms.

|      |         |         |               | Exports  |                                  |
|------|---------|---------|---------------|----------|----------------------------------|
|      |         | (1)     | (2)<br>Grants | (3)      | (4) (5) (6)<br>Matchmaking       |
| Year | Treated | Mean    | Median        | Nb firms | Mean Median Nb firms             |
| 2012 | No      | 2264.89 | 88.19         | 17559    | 1212.01228.863729715.8296.947    |
| 2012 | Yes     | 3727.61 | 318.55        | 91       |                                  |
| 2013 | No      | 2252.67 | 84.05         | 18456    | 1214.88 233.87 3814              |
| 2013 | Yes     | 3612.97 | 878.04        | 192      | 915.03 148.36 69                 |
| 2014 | No      | 2195.55 | 80.41         | 18385    | 1202.20 237.10 3833              |
| 2014 | Yes     | 3127.52 | 676.02        | 199      | 775.51 176.76 113                |
| 2015 | No      | 2324.11 | 79.77         | 18347    | 1267.43219.894021650.08170.93183 |
| 2015 | Yes     | 3124.48 | 514.08        | 321      |                                  |
| 2016 | No      | 2385.09 | 84.26         | 17540    | 1257.88234.964118715.70179.17220 |
| 2016 | Yes     | 2689.75 | 472.48        | 461      |                                  |
| 2017 | No      | 2598.39 | 92.67         | 17754    | 1345.66231.644255587.12157.32270 |
| 2017 | Yes     | 2704.60 | 558.40        | 512      |                                  |
| 2018 | No      | 2517.46 | 77.43         | 18797    | 1333.67224.014395613.04193.87294 |
| 2018 | Yes     | 2668.13 | 461.19        | 579      |                                  |
| 2019 | No      | 2554.60 | 67.10         | 18952    | 1347.59235.044392697.22183.36340 |
| 2019 | Yes     | 2823.46 | 436.69        | 652      |                                  |
| 2020 | No      | 2205.35 | 71.91         | 17704    | 1125.37213.374451776.82182.76368 |
| 2020 | Yes     | 2609.99 | 435.13        | 696      |                                  |
| 2021 | No      | 2027.00 | 60.11         | 11595    | 1054.55153.453140837.62188.37406 |
| 2021 | Yes     | 3191.51 | 535.79        | 839      |                                  |

Table 1. Summary statistics for export values (thousand euros, at 2012 prices) Note: Export values are in thousands of euros and 2012 values.

Firms receiving grants are typically larger and export more than those that do not. Matchmaking services in international markets can often be purchased privately, making them more accessible to larger companies that can afford to customize these services for their specific needs. In contrast, smaller firms frequently rely on AICEP, which provides matchmaking services free of charge. These differences are also reflected in labor productivity. Table B.1 shows that firms receiving grants are more productive than their counterparts, whereas, for matchmaking services, the median labor productivity is similar between treated and untreated firms.

Finally, regarding the sectoral distribution of firms benefiting from grants and matchmaking services in France, Table B.2 in Appendix B ranks the top five NACE 2-digit sectors. In both export promotion policies, wholesale international traders and firms operating in the manufacture of wearing apparel and textiles dominate.

#### 4. Methodology

#### 4.1. Identification strategy

The methodology to measure the impact of grants and matchmaking services is similar. The model specification of the event study is described by the following specification:

$$y_{i,t} = \gamma_i + \lambda_t + \zeta_{sec} + \delta Di D_{i,t} + \beta X_{i,t} + \varepsilon_{i,t}.$$
 (1)

where  $y_{i,t}$  is the logarithm of exports of firm *i* in year *t*,  $\gamma_i$  is a firm fixed effect,  $\lambda_t$  a time fixed effect, and  $\zeta_{sec}$  a sector fixed effect.  $X_{i,t}$  is a matrix of controls with  $\beta$  a vector of coefficients and  $\varepsilon_{i,t}$  represents the unobserved shock that is assumed to be uncorrelated with the treatment,  $DiD_{i,t}$ .

The control variables in  $X_{i,t}$  are the natural logarithm of GVA per employee and the firm's size class, defined along four categories: micro, small, medium, and large, according to the classification of the European Commission.<sup>5</sup> The absorbing treatment indicator is a dummy taking value one from the year the firm receives the support to export (support<sub>t</sub>) onwards, i.e.  $DiD_{i,t} = 1[t \ge support_t]$ . The primary coefficient of the model ( $\delta$ ) indicates the average treatment effect on treated (ATT), and we cluster the standard errors at the firm level.

We estimate the equation 1 using the interaction-weighted (IW) estimator proposed by Sun and Abraham (2021). This estimator allows the treatment effect to differ according to the moment of treatment. Moreover, it corrects for the heterogeneity of firm's responses to AICEP's support.<sup>6</sup> The ATT is estimated using the never-treated firms as counterfactual.

Although the (Sun and Abraham 2021) estimator is robust, it requires two assumptions. Firstly, it requires the presence of parallel trends, i.e., treated and untreated firms need to trend together before intervention. We check the validity of this assumption on our dataset by plotting nine years of pre-treatment coefficients. The second assumption is the presence of no anticipation behavior prior to treatment. According to our specifications, this means that firms did not try to increase their exports to France before AICEP's matchmaking service, or they were not successful in increasing exports before grants were received. Additionally, a placebo test is considered to check whether the increase in firm's exports was caused by the support or was due to other determinants. The next section details these robustness tests.

<sup>5.</sup> For more information on the classification criteriua, see link.

<sup>6.</sup> See also Baker et al. (2022) for a discussion on this estimator.

#### 4.2. Robustness

4.2.1. **Parallel trends test**. One of the main prerequisites for the use of the DiD method is the assumption of a parallel trend, i.e., the treatment and control groups trend similarly before the intervention. According to the literature, such as Nunn and Qian (2011); Gu *et al.* (2021); Díaz (2024), we adopt the event study method to test the parallel trends assumption. Specifically, we rely on a test based on a model adapted from equation 1:

$$y_{i,t} = \gamma_i + \lambda_t + \zeta_{sec} + \sum_{\tau=-5}^{-2} \varphi_{\tau} Di D_{i,t}^{\tau} + \sum_{\tau=0}^{5} \delta_{\tau} Di D_{i,t}^{\tau} + X_{i,t}\beta + \varepsilon_{i,t}$$

$$(2)$$

where  $y_{i,t}$  is the natural logarithm of exports of firm i in year t,  $\gamma_i$  is the firm fixed effect,  $\lambda_t$  a time fixed effect,  $\zeta_{sec}$  is a sector fixed effect, and  $X_{i,t}$  is the set of controls (the same as in equation 2), with coefficients  $\beta$ .  $\varepsilon_{i,t}$  represents the unobserved shock that is assumed to be uncorrelated with the treatment,  $DiD_{i,t}^{\tau}$ , and  $\varphi_{\tau}$  and  $\delta_{\tau}$ , are the coefficients of  $DiD_{i,t}^{\tau}$  before and after the treatment. Observations of 5 or more years before and after the intervention are grouped.  $\varphi_{\tau}$  and  $\delta_{\tau}$  indicate if the groups were trending equally before and after the intervention. The mechanics of the event study's parallel trend test is: if  $\varphi_{\tau}$  ( $\delta_{\tau}$ ) is not significantly different from zero, then the two groups were trending the same before (after) the intervention.

4.2.2. **Falsification test**. To check the robustness of the DiD estimates, we implement the falsification test along the line of Li *et al.* (2021) and Ai *et al.* (2022), which change the treatment date, i.e., the implementation of the exogenous shock. We set the treatment to 1, 2 and 3 years before the real intervention date and the estimated model is similar to our primary specification in equation 1, as follows:

$$y_{i,t} = \gamma_i + \lambda_t + \zeta_{sec} + \sum_{\tau=-9}^{-2} \varphi_{\tau}^* fakeDiD_{i,t}^{\tau} + \sum_{\tau=0}^{8} \delta_{\tau}^* fakeDiD_{i,t}^{\tau} + X_{i,t}\beta + \varepsilon_{i,t},$$
(3)

where the  $fakeDiD_{i,t}^{\tau}$  refers to the treated group with fake intervention time. If the estimated coefficient of  $fakeDiD_{i,t}^{\tau}$ ,  $\hat{\delta}_{\tau}^{*}$ , is significant, then some external shock before the intervention affected exports of firms in the treatment group or an anticipatory behavior could be present.

4.2.3. **Placebo test**. Finally, we perform the placebo test to assess whether the increase in firms' exports comes from the support rather than from other external

factors, by randomly assigning firms to control and treatment groups, with the same proportion of our real dataset. Then, again, we randomly generate the treatment year when the AICEP support was given. The estimated test model is also similar to our primary specification in equation 1:

$$y_{i,t} = \gamma_i + \lambda_t + \zeta_{sec} + \sum_{\tau=-9}^{-2} \varphi^{\tau} fakeDiD_{i,t}^{*\tau} + \sum_{\tau=0}^{8} \delta^{\tau} fakeDiD_{i,t}^{*\tau} + X_{i,t}\beta + \varepsilon_{i,t},$$
(4)

where the  $fakeDiD_{i,t}^{*\tau}$  refers to the randomly assigned treated group with the respective randomly assigned intervention year.

In this placebo test, after generating random intervention dates and computing the ATT through our event study model, we expect the estimated ATT to be normally distributed around zero and non-statistically significant.

#### 5. Results

#### 5.1. Global impact on exports

The global impact of AICEP's grants and matchmaking services on the exports of Portuguese firms, as obtained from Equation 1, are presented in Table 2.

|   |                     | In of exports                     |                      |                                   |  |  |  |  |
|---|---------------------|-----------------------------------|----------------------|-----------------------------------|--|--|--|--|
|   | (1)                 | (2)                               | (3)                  | (4)                               |  |  |  |  |
|   | Gr                  | rants                             | Matchmaking          |                                   |  |  |  |  |
| DiD                                     | 0.1019*<br>(0.0542) | 0.1226 <sup>***</sup><br>(0.0436) | 0.2044**<br>(0.1030) | 0.2749 <sup>***</sup><br>(0.0905) |  |  |  |  |
| Controls                                | No                  | Yes                               | No                   | Yes                               |  |  |  |  |
| Adjusted R <sup>2</sup><br>Observations | 0.8139<br>166,377   | 0.8716<br>166,377                 | 0.7821<br>37,934     | 0.8123<br>37,934                  |  |  |  |  |

#### Table 2. Global impact on firms' exports

Note: The controls the logarithm of the gross value added per employee, sector at three digit NACE Rev.2 class and firms' size class. The estimation also included firm and year fixed effects. The standard errors are clustered at the firm level. \*\*\*, \*\*, and \*, indicate the 1%, 5%, and 10% significance levels, respectively.

Columns (1) and (2) in table 2 refer to grants and columns (3) and (4) refer to matchmaking, with each pair presenting results without and with controls,

respectively. The coefficient related to DiD in Table 2 represents the primary coefficient in our event study model - the estimated ATT.

For grants, the estimated ATT without controls (Column 1) is statistically significant at the 10% level. However, when controls for GVA per employee, sector and firm size class are included (Column 2), the significance increases to the 1% level. Receiving a grant resulted in a 13% increase in treated firms' total exports compared to those that did not receive financial support ( $(e^{0.1226} - 1) \times 100$ ). Similarly, for matchmaking, the estimated ATT is significant at the 5% level without controls (Column 3) and at the 1% level with controls (Column 4). With controls, this service led to a 31.6% increase in firms' exports to France.

Several studies have also analyzed the impact of EPAs on firms' exports, reporting substantial effects. Van Biesebroeck *et al.* (2015) investigated export promotion policies in Canada and found an increase in exports of approximately 16%. Using a DiD estimator in the context of the global financial crisis recovery, Van Biesebroeck *et al.* (2016) reported effects ranging from 19.5% to 24.2% in Belgium and from 13.4% to 22.5% in Peru. Additionally, Martincus and Carballo (2010) analyzed a set of services provided by Peru's EPA and observed an increase in exports of 28.5%.

Focusing on matchmaking services, Makioka (2021) analyzed the Japanese EPA's support for firms attending trade fairs, finding a 25% increase in exports during the year following the fair. Other studies have estimated more modest effects of EPAs' matchmaking services on firms' export values. For instance, Munch and Schaur (2018) reported a 5.8% increase two years after the support, while Broocks and Van Biesebroeck (2017) found an effect of 14.4%.

It is worth noting that these studies typically estimate the impact on firms' total exports after benefiting from matchmaking services. In contrast, our approach provides a more accurate estimation of the effect of matchmaking because it considers both the specific markets for which firms received information and the destinations of their exports.

#### 5.2. Heterogeneous impact on exports by firm size and sector

The empirical literature demonstrates that the effects of Export Promotion Programs (EPPs) can be heterogeneous, varying by firm size and sector of activity (Martincus and Carballo 2010; Cruz 2014; Broocks and Van Biesebroeck 2017; Munch and Schaur 2018). Based on standard international trade theory (Melitz 2003) and the empirical works summarized in Section 2.1, we anticipate that EPPs are more effective for smaller firms. These firms are generally less productive and possess fewer assets to use as collateral, making it more challenging for them to cover the sunk costs required to access international markets (Wagner 2007).

Regarding sectors, certain characteristics are likely to drive stronger impacts from EPPs. Using the IES dataset, we identified firms' sectors according to the 3-digit Statistical Classification of Economic Activities (Carré 2008) and conducted our event study for the five sectors receiving the most support in grants and

matchmaking, as shown in Table B.2. Table 3 reports estimates for sectors where a significant ATT was observed: sector 463 ("Wholesale of food, beverages, and tobacco") and sector 141 ("Manufacture of wearing apparel") for grants, and sector 139 ("Manufacture of other textiles") for matchmaking.

In general, the disaggregated results by size class and sector, presented in Table 3, exhibit greater variability compared to the global effects reported in Table 2. This was expected, as these results are based on subsamples of the main dataset. For grants targeting internationalization, the impact on exports was statistically significant only for micro and small firms. The estimated DiD coefficient in column (1) shows that this type of support increased the total exports of micro and small firms by 18.1% compared to those that did not receive grants.

In terms of sectoral impacts, we identified significant effects in "Wholesale of household goods" (column 3) and "Manufacture of wearing apparel" (column 4). The former experienced a 35.6% increase in total exports, while the latter saw an increase of approximately 47% compared to the control group.

For matchmaking in France, the estimated DiD coefficient in column (5) indicates that micro and small firms achieved a 39.8% increase in their exports to France. Column (6) reveals a significant effect for medium and large firms, albeit only at the 10% significance level, with an export increase of approximately 23.2%. We also observed heterogeneous sectoral impacts. The coefficient was statistically significant only for the sector "Manufacture of other textiles." The model estimates that the matchmaking service increased exports of firms in this sector to France by 136.5% compared to firms that did not receive the treatment.

|  | (1)                          | (2)                         | In of (<br>(3)                 | exports<br>(4)             |      | (5)                         | (6)                         |      | (7)                        |
|--|------------------------------|-----------------------------|--------------------------------|----------------------------|------|-----------------------------|-----------------------------|------|----------------------------|
|  | Grants                       |                             |                                |                            |      |                             | Matchmal                    | king |                            |
|  | Size Sector                  |                             |                                | ctor                       | Size |                             |                             |      | Sector                     |
|  | Micro<br>Small               | Medium<br>Large             | NACE<br>463                    | NACE<br>141                |      | Micro<br>Small              | Mid<br>Large                |      | NACE<br>139                |
| DiD  | 0.167***<br>(0.056)          | 0.010<br>(0.049)            | 0.305 <sup>**</sup><br>(0.138) | 0.389**<br>(0.169)         |      | 0.335**<br>(0.150)          | 0.21*<br>(0.113)            |      | 0.862***<br>(0.216)        |
| Sec. FE<br>Size FE<br>Adj R <sup>2</sup><br>Obs. | Yes<br>No<br>0.84<br>135,693 | Yes<br>No<br>0.90<br>30,683 | No<br>Yes<br>0.90<br>6,979     | No<br>Yes<br>0.87<br>4,055 |      | Yes<br>No<br>0.78<br>21,572 | Yes<br>No<br>0.84<br>16,361 |      | No<br>Yes<br>0.85<br>1,316 |

Note: The table reports the results of Equation 2 for size classes (micro-small and mediumlarge) and the main sectors in which supported firms operate (NACE 463 - "Wholesale of household goods," 141 - "Manufacture of wearing apparel," and 139 - "Manufacture of other textiles"). The model controls for the natural logarithm of GVA per employee. The estimations also include firm and time fixed effects. Standard errors are clustered at the firm level. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

Table 3. Results by firm size class and sector of activity

#### 6. Robustness of results

In this section, we conduct a series of tests on our estimated event study models. The analysis begins with an assessment of the robustness of the estimated global effects, followed by tests evaluating the robustness of the estimated impacts across firms' size classes and sectors of activity. Specifically, we test the parallel trend assumption and perform falsification and anticipation tests to validate our findings.

#### 6.1. Global effect

As discussed in Subsection 4.2.1, the parallel trend hypothesis is central to the DiD methodology. The parallel trends between the treatment and control groups are reflected in the coefficients of the dummy treatment variables,  $\varphi_{\tau}$  and  $\delta_{\tau}$ . Figure 1 presents the estimated DiD coefficients for grants and matchmaking.

In both cases, prior to the policy intervention, the control and treatment groups exhibit parallel trends, with one exception for matchmaking two years before the intervention. This deviation could suggest isolated anticipatory behavior by treated firms. The falsification test provides additional evidence on this issue.

After the period when grants are received, all estimated treatment coefficients  $(\delta_{\tau})$  are positive. Furthermore, the financial support appears to impact firms' exports with a delay, as the estimated coefficients only become significantly different from zero three years after the financial support is provided.

For matchmaking, the treatment coefficient  $\delta_{\tau}$  is not significantly different from zero during the intervention period. However, it subsequently increases systematically, remaining positive and significant. This suggests that the matchmaking service enhanced firms' exports compared to those that did not receive the support.

On the other hand, the falsification test described in Subsection 4.2.2 involves shifting the treatment period to 1, 2, and 3 years before the actual date. This sensitivity analysis assesses whether external factors influenced the increase in exports for the treatment group relative to the control group prior to the actual treatment date. The results in Table 4 show that the false ATTs are significant for both types of support when the intervention dates are shifted to one year before the actual dates, albeit only at the 10% significance level. However, these significant effects disappear when the intervention dates are shifted to 2 and 3 years before the actual support dates.

As for the placebo test discussed in Subsection 4.2.3, we investigate whether the significant effect on firms' exports stems from these interventions rather than external factors. We ran Equation 4 on the data with fake interventions 1,000 times and present the results in Figure C.1 (the 1,000 estimated  $\delta_{\tau}^*$  values are reported for grants in the top panel and matchmaking in the bottom panel). The mean of the estimated coefficients, indicated by the vertical dashed line, is close to zero, while the estimated values from our event study model are 0.123 and 0.275 for grants and matchmaking, respectively. Furthermore, almost all estimates are not





(b) Matchmaking

#### Figure 1: Parallel trend test

Note: The figure reports the parallel trend test, i.e., the core coefficients of our event study model, from Equation 2:  $\varphi_{\tau}$ , representing coefficients before the support, and  $\delta_{\tau}$ , for coefficients after the support. The zero on the horizontal axis represents the time of the support. We set the period just before the support as the reference, i.e., set to zero. The vertical axis represents the grant's estimated coefficients and confidence intervals.

|   |                    | In of exports         |                    |  |                           |                    |                   |  |  |
|---|--------------------|-----------------------|--------------------|--|---------------------------|--------------------|-------------------|--|--|
|   | (1)                | (1) (2) (3)<br>Grants |                    |  | (4) (5) (6<br>Matchmaking |                    |                   |  |  |
|   | 1 year             | 2 years               | 3 years            |  | 1 year                    | 2 years            | 3 years           |  |  |
| fake DiD                                | 0.153*<br>(0.0814) | 0.0318<br>(0.074)     | 0.0231<br>(0.0668) |  | 0.1421*<br>(0.0837)       | 0.1083<br>(0.0941) | 0.0385<br>(0.094) |  |  |
| Adjusted R <sup>2</sup><br>Observations | 0.8934<br>166,377  | 0.8934<br>166,377     | 0.8934<br>166,377  |  | 0.81427<br>37,934         | 0.81425<br>37,934  | 0.81421<br>37,934 |  |  |

#### Table 4. Anticipatory effect

The table reports the results of Equation 3, where the actual support date is anticipated. Columns (1) to (3) pertain to financial grants, while columns (4) to (6) relate to matchmaking. The controls include the natural logarithm of GVA per employee, sector classification at the three-digit RACE Rev 2 level, and firm size class. The estimates also account for firm and year fixed effects. Standard errors are clustered at the firm level. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

significant at the 5% level. These results suggest that the increase in firms' exports is attributable to the intervention rather than other external factors.

#### 6.2. Heterogeneous effect by firm size and sector

Similarly to what we do in the model for the global effect in Subsection 4.2.1, we perform the parallel trends tests in the context of heterogeneous effects for firm size classes and sector of activity. Figure C.2 report the results. The global patterns discussed in the previous section appear to mainly reflect the results of micro and small firms, which represent 81.6 percent and 56.9 percent of the sample for grants and matchmaking services, respectively. Moreover, these global patterns are very close to those exhibited by the three selected sectors under analysis. In general, the estimated  $\varphi_{\tau}$  mostly fluctuate around zero before the intervention point, i.e., the control and treatment groups trended together before receiving the grants and the matchmaking services. A single exception to this behavior is observed for grants to micro and small firms. The significant effect two years before the intervention likely explains the significance of the estimate for this period in the global analysis of the previous section.

Results by dimensional classes suggest that effects on exports are only significant for micro and small firms. For these firms, the effect of the intervention for grants becomes significant at the third year after the support is received, displaying a systematic increasing magnitude afterwards. Regarding the matchmaking service, it only became relevant for these firms at the fifth and following years in analysis. Medium and large firms do not display significant benefits from the AICEP's intervention. In addition, the plot on the left bottom of Figure C.2 shows that the firms in sectors "Wholesale of food, beverages and tobacco" (463) and "Manufacturing of clothing" (141) had an increase in their total exports one year after receiving grants. This effect became significant two years after the intervention, one year later than in the global results, but then lost significance for one year and returned to the significance status in the fourth year for both sectors and in fifth and following years for the latter sector. For sector "Manufacture of other textiles" (139) the effect from the matchmaking service mimics the global pattern as reported in the plot on the right bottom of Figure C.2. The impact is observed immediately after treatment, that is, the coefficients  $\delta_{\tau}$  begin to increase rapidly and systematically after the intervention and remain high afterwards.

Similarly to subsection 4.2.2, where we present the falsification test for the global effect of grants and matchmaking by shifting the treatment period to 1, 2, and 3 years before the actual date, we now perform the test in the heterogeneous effect setup. Results for grants are reported in Table C.1 and for matchmaking in Table C.2. In all cases, results indicate that when we replace the real intervention dates with the falsified ones, the treatment does not significantly affect exports at a 5% significance level. Placebo tests were also performed in the framework of this heterogeneous approach, however, the results mimic those of the global scenario described in the previous section. Therefore, they are not discussed in the paper, but they are available from the authors upon request.

#### 7. Conclusion

The paper analyses the impact of financial grants for internationalization and matchmaking services in international markets on exports by Portuguese firms. To evaluate the impact of financial grants we consider all firms that exported in from 2012 to 2021, while the impact of matchmaking is focused on exports to France in the same period. Our empirical analysis uses a rich and unique dataset built from three different sources: the AICEP clients' dataset, the official trade transaction dataset, and the official business register dataset.

Results from our event study indicate a sizable impact from these export promotion policies. Grants for internationalization led to an increase in total exports of about 12%, reaching almost 17% for micro and small firms. In sectors "Wholesale of household goods" and "Manufacture of wearing apparel", the impacts were about 34% and 47%, respectively. As for the effect of matchmaking in France, the increase in firms' exports to France was about 29%. Micro and small firms recorded an increase in their exports by about 38%, and for firms in the "Manufacture of other textiles" the matchmaking led to an increase of 137% in exports to that market.

Although results are sizable and robust, it is necessary to acknowledge that the assumption that the treatment given by the Portuguese EPA is not correlated to any variable captured by our error term remains a challenge. In fact, companies that approach AICEP can have more information about their services, be more productive, less risk averse, and more willing to venture into international markets. This could lead to a bias in the estimated effect of the EPPs but would hardly eliminate it.

Looking ahead, the research agenda includes additional exercises, that should shore up and expand results. Firstly, it would be interesting to assess the differential impact of receiving the two treatments versus just one. Such weak versus strong treatment analysis would further reduce concerns raised in the previous paragraph. Secondly, future work may go beyond the intensive margin of exports, thus investigating the long-term effect of EPA's services on firms' export prices, product quality, markups, employment and productivity. Thirdly, it may be relevant to try a partial cost benefit analysis by comparing the cost of policies with a proxy of the value added generated by the new exports.

There is extensive literature documenting the benefits of having an outwardoriented economy. As previously discussed, the primary advantage is a higher productivity for domestic firms, which stems from the interaction with foreign companies possessing more advanced (or distinct) knowledge bases.<sup>7</sup> In recent years, however, the global economy has been shifting towards a more protectionist stance. This trend towards higher tariffs and non-tariff barriers implemented by major economic blocks, along with the usual retaliatory actions, risks severely

<sup>7.</sup> See the case of *learning-by-exporting* (De Loecker 2007).

reducing social welfare and global stability. In contrast, the adoption of export promotion policies, primarily aimed at correcting market failures, stands as a good policy approach, with significant positive impacts on firms' exports.

#### References

- Aalto, Eero and Robin Gustafsson (2020). "Export Promotion Rationales and Impacts – A Review." ETLA Reports 100, The Research Institute of the Finnish Economy.
- Ai, Hongshan, Tenglong Zhong, and Zhengqing Zhou (2022). "The real economic costs of COVID-19: Insights from electricity consumption data in Hunan Province, China." *Energy Economics*, 105, 105747.
- Amador, João and Luca Opromolla (2013). "Product and destination mix in export markets." *Review of World Economics (Weltwirtschaftliches Archiv)*, 149(1), 23– 53.
- Amador, João and Luca Opromolla (2017). "Trade Margins and. Cohorts of Traders in Portugal." *Economic Studies of Banco de Portugal*, 5, 1–18.
- Atkin, David, Amit K Khandelwal, and Adam Osman (2017). "Exporting and firm performance: Evidence from a randomized experiment." *The Quarterly Journal of Economics*, 132(2), 551–615.
- Baker, Andrew C., David F. Larcker, and Charles C.Y. Wang (2022). "How much should we trust staggered difference-in-differences estimates?" *Journal of Financial Economics*, 144(2), 370–395.
- Banerjee, Abhijit Vinayak (2013). "Microcredit under the microscope: What have we learned in the past two decades, and what do we need to know?" Annual Review of Economics, 5(1), 487–519.
- Bernard, Andrew B and J Bradford Jensen (2004). "Why some firms export." *Review of Economics and Statistics*, 86(2), 561–569.
- Broocks, Annette and Johannes Van Biesebroeck (2017). "The impact of export promotion on export market entry." *Journal of International Economics*, 107, 19–33.
- Buus, Magnus T., Jakob R. Munch, Joel Rodrigue, and Georg Schaur (2022). "Do Export Support Programs Affect Prices, Quality, Markups and Marginal Costs? Evidence from a Natural Policy Experiment." *The Review of Economics and Statistics*, pp. 1–45.
- Carré, H (2008). "NACE Rev. 2, Statistical classification of economic activities in the European Community." *General and Regional Statistics*.
- Castela, Tiago Pereira Leite (2021). A Internacionalização de Empresas na perspetiva da Diplomacia Económica. Ph.D. thesis, ISCTE-Instituto Universitario de Lisboa (Portugal).
- Comi, Simona and Laura Resmini (2020). "Are export promotion programs effective in promoting the internalization of SMEs?" *Economia Politica*, 37, 547–581.
- Copeland, Brian R (2007). "Is there a case for trade and investment promotion policy." *Trade Policy Research*, pp. 1–64.
- Cruz, Marcio (2014). "Do export promotion agencies promote new exporters?" World Bank Policy Research Working Paper, (7004).
- Cruz, Marcio, Daniel Lederman, Laura Zoratto, et al. (2018). "The anatomy and the impact of export promotion agencies." Research Handbook on Economic

Diplomacy. Cheltenham: Edward Elgar Publishing Limited, pp. 94–108.

- de Chaisemartin, Clément and Xavier D'Haultfœuille (2020). "Two-Way Fixed Effects Estimators with Heterogeneous Treatment Effects." American Economic Review, 110(9), 2964–96.
- De Loecker, Jan (2007). "Do exports generate higher productivity? Evidence from Slovenia." *Journal of International Economics*, 73(1), 69–98.
- Díaz, Julián P (2024). "Does dollarization promote trade? Evidence from two recent episodes." *Applied Economics*, 56(17), 2058–2076.
- Gil-Pareja, Salvador, Rafael Llorca-Vivero, José Antonio Martínez-Serrano, and Francisco Requena-Silvente (2015). "Regional export promotion offices and trade margins." *Review of World Economics*, 151, 145–167.
- Gu, Yizhen, Chang Jiang, Junfu Zhang, and Ben Zou (2021). "Subways and road congestion." *American Economic Journal: Applied Economics*, 13(2), 83–115.
- Harrison, Ann and Andrés Rodríguez-Clare (2010). "Trade, foreign investment, and industrial policy for developing countries." *Handbook of Development Economics*, 5, 4039–4214.
- Hayakawa, Kazunobu, Hyun-Hoon Lee, and Donghyun Park (2014). "Do export promotion agencies increase exports?" *The Developing Economies*, 52(3), 241–261.
- Head, Keith and John Ries (2010). "Do trade missions increase trade?" *Canadian Journal of Economics/Revue Canadienne d'Économique*, 43(3), 754–775.
- Li, Ke, Weihong Yuan, Jianglong Li, and Hongshan Ai (2021). "Effects of time-dependent environmental regulations on air pollution: Evidence from the Changsha-Zhuzhou-Xiangtan region, China." World Development, 138, 105267.
- Lileeva, Alla and Daniel Trefler (2010). "Improved access to foreign markets raises plant-level productivity... for some plants." *The Quarterly Journal of Economics*, 125(3), 1051–1099.
- Makioka, Ryo (2021). "The impact of export promotion with matchmaking on exports and service outsourcing." *Review of International Economics*, 29(5), 1418–1450.
- Martincus, Christian Volpe and Jerónimo Carballo (2010). "Beyond the average effects: The distributional impacts of export promotion programs in developing countries." *Journal of Development Economics*, 92(2), 201–214.
- Melitz, Marc J (2003). "The impact of trade on intra-industry reallocations and aggregate industry productivity." *Econometrica*, 71(6), 1695–1725.
- Melitz, Marc J and Stephen J Redding (2014). "Heterogeneous firms and trade." *Handbook of International Economics*, 4, 1–54.
- Munch, Jakob and Georg Schaur (2018). "The effect of export promotion on firmlevel performance." *American Economic Journal: Economic Policy*, 10(1), 357–387.
- Nunn, Nathan and Nancy Qian (2011). "The potato's contribution to population and urbanization: evidence from a historical experiment." *The Quarterly Journal of Economics*, 126(2), 593–650.

- OCDE (2018). "Mapping of Investment Promotion Agencies in OECD Countries." OCDE working paper.
- Roberts, Mark J and James R Tybout (1997). "The decision to export in Colombia: An empirical model of entry with sunk costs." *The American Economic Review*, pp. 545–564.
- Rose, Andrew K (2007). "The foreign service and foreign trade: embassies as export promotion." *World Economy*, 30(1), 22–38.
- Srhoj, Stjepan, Vanja Vitezić, and Joachim Wagner (2023). "Export Boosting Policies and Firm Performance: Review of Empirical Evidence Around the World." *Jahrbücher für Nationalökonomie und Statistik*.
- Srhoj, Stjepan and Janette Walde (2020). "Getting ready for EU Single Market: The effect of export-oriented grant schemes on firm performance." *Structural Change and Economic Dynamics*, 52, 279–293.
- Sun, Liyang and Sarah Abraham (2021). "Estimating dynamic treatment effects in event studies with heterogeneous treatment effects." *Journal of Econometrics*, 225(2), 175–199.
- Van Biesebroeck, Johannes, Jozef Konings, and Christian Volpe Martincus (2016). "Did export promotion help firms weather the crisis?" *Economic Policy*, 31(88), 653–702.
- Van Biesebroeck, Johannes, Emily Yu, and Shenjie Chen (2015). "The impact of trade promotion services on Canadian exporter performance." *Canadian Journal of Economics/Revue Canadienne d'Économique*, 48(4), 1481–1512.
- Wagner, Joachim (2007). "Exports and productivity: A survey of the evidence from firm-level data." *World Economy*, 30(1), 60–82.





Figure A.1: Exports of treatment and control firms - grants Note: Vertical dashed lines identify the mean values and exports are in euros of 2012.



Figure A.2: Exports of treatment and control firms - matchmaking Note: Vertical dashed lines identify the mean values and the exports are in euros of 2012.

#### Appendix B: Descriptive statistics

|           | Grants            |                    |                   |  | Matchmaking      |                  |                   |  |  |
|-----------|-------------------|--------------------|-------------------|--|------------------|------------------|-------------------|--|--|
| _         | Mean              | Median             | Std dev           |  | Mean             | Median           | Std dev           |  |  |
| No<br>Yes | 16,315<br>194,939 | 106,442<br>125,779 | 361,363<br>66,541 |  | 45,399<br>36,058 | 24,036<br>24,204 | 165,017<br>64,537 |  |  |

Table B.1. Gross value added per employee

.

Note: "No" and "Yes" refer to control and treatment firms, respectively.

| Rank        | NACE sector  |
|-------------|--|
| Grants      |  |
| 1           | 464 - Wholesale of household goods, except food, beverages and tobacco |
| 2           | 463 - Wholesale of food, beverages and tobacco                         |
| 3           | 467 - Other specialized wholesale                                      |
| 4           | 469 - Non-specialized wholesale trade                                  |
| 5           | 141 - Manufacture of wearing apparel, except fur apparel               |
| Matchmaking |  |
| 1           | 141 - Manufacture of wearing apparel, except fur apparel               |
| 2           | 464 - Wholesale of household goods, except food, beverages and tobacco |
| 3           | 139 - Manufacture of other textiles                                    |
| 4           | 463 - Wholesale of food, beverages and tobacco                         |
| 5           | 257 - Manufacture of cutlery, tools and general hardware               |

Table B.2. Top 5 sectors in treatment group

Note: The firm's sectors are sourced in the INE dataset. We consider the 3-digit Statistical Classification of Economic Activities in the European Union (Carré 2008) as the sector.



Appendix C: Robustness of results

Figure C.1: Placebo test: Grants and matchmaking

Note: This figure reports the 1000 estimations of the DiD coefficients from equation 4. The vertical black line is the estimation from our primary Equation 2, the dashed horizontal line is the average p-value from this simulation, and the red horizontal line refers to the 5% significance level.





Note: The figure reports the parallel trend test, i.e., the core coefficients of our event study model, from Equation 2:  $\varphi_{\tau}$ , representing coefficients before the support, and  $\delta_{\tau}$ , for coefficients after the support. The zero on the horizontal axis represents the time of the support. We set the period just before the support as the reference, i.e., set to zero. The vertical axis represents the estimated coefficients for grants (a) and matchmaking (b) and their confidence intervals.

|              | (9)                | 3 years | 0.0208       | (0.0963) | 0.91219                 | 6,008        |             |
|--------------|--------------------|---------|--------------|----------|-------------------------|--------------|-------------|
|              | (8)<br>NACE 14:    | 2 years | 0.1175       | (0.1263) | 0.91218                 | 6,008        |             |
|              | (2)                | 1 year  | 0.1052       | (0.1045) | 0.91222                 | 6,008        |             |
| ts           | (9)                | 3 years | 0.1607       | (0.1384) | 0.8986                  | 6,978        |             |
| In of export | (5)<br>NACE 463    | 2 years | $0.1796^{*}$ | (0.0973) | 0.8986                  | 6,978        |             |
|              | (4)                | 1 year  | $0.3561^{*}$ | (0.1935) | 0.8985                  | 6,978        |             |
|              | (3)<br>nall        | 3 years | 0.0914       | (0.0607) | 0.8438                  | 135,693      |             |
|              | (2)<br>icro and Sn | 2 years | 0.0941       | (0.0717) | 0.84386                 | 135,693      |             |
|              | (1)<br>Mi          | 1 year  | $0.1205^{*}$ | (0.0665) | 0.84388                 | 135,693      | (<br>)<br>) |
|              |                    |         | Fake DiD     |          | Adjusted $\mathbb{R}^2$ | Observations |             |

| Grants       |
|--------------|
| effect:      |
| Anticipatory |
| C.1.         |
| Table        |

Note: The table reports the results of equation 3, where we anticipate the actual support date. Columns (1) to (3) report financial grants and columns (4) to (6) refer to the matchmaking. We control for the ln of the GVA per employee, sector at 3 digits NACE Rev. 2 level and firms' size class. The estimates include firm and year fixed effects. Standard errors are clustered at the firm level. \*\*\*, \*\* and \*, indicate the significance levels of 1%, 5%, and 10%, respectively.

|   |                     | In of exports                  |                    |  |                    |                    |                    |  |  |
|---|---------------------|--------------------------------|--------------------|--|--------------------|--------------------|--------------------|--|--|
|   | (1)<br>Mi           | (1) (2) (3)<br>Micro and Small |                    |  | (4)                | (5)<br>NACE 139    | (6)                |  |  |
|   | 1 year              | 2 years                        | 3 years            |  | 1 year             | 2 years            | 3 years            |  |  |
| Fake DiD                                | 0.2085*<br>(0.1124) | 0.1664<br>(0.180)              | 0.0036<br>(0.1567) |  | 0.0278<br>(0.2387) | 0.0378<br>(0.1119) | 0.0474<br>(0.0837) |  |  |
| Adjusted R <sup>2</sup><br>Observations | 0.77844<br>24,071   | 0.77851<br>24,071              | 0.77851<br>24,071  |  | 0.8238<br>766      | 0.8238<br>766      | 0.8238<br>766      |  |  |

| Table C | 2Δ    | nticinator | v effect. | Matchma     | king  |
|---------|-------|------------|-----------|-------------|-------|
| Table C | .Z. A | nticipator | y enect.  | IVIalCIIIIa | KIIIg |

Note: The table reports the results of equation 3, where we anticipate the actual support date. Columns (1) to (3) report the financial grants, and columns (4) to (6) refer to the matchmaking. We control for the ln of the GVA per employee, sector at 3 digits NACE Rev. 2 level and firms' size class. The estimates include firm and year fixed effects. Standard errors are clustered at the firm level. \*\*\*, \*\*, and \*, indicate the 1%, 5%, and 10% significance levels, respectively.

## Working Papers

## 2022

- 1|22 Business cycle clocks: Time to get circular Nuno Lourenço | António Rua
- 2 | 22 The Augmented Bank Balance-Sheet Channel of Monetary Policy Christian Bittner | Diana Bonfim | Florian

Heider | Farzad Saidi | Glenn Schepens | Carla Soares

3|22 Optimal cooperative taxation in the global economy

V. V. Chari | Juan Pablo Nicolini | Pedro Teles

- 4|22 How Bad Can Financial Crises Be? A GDP Tail Risk Assessment for Portugal Ivan De Lorenzo Buratta | Marina Feliciano | Duarte Maia
- 5|22 Comparing estimated structural models of different complexities: What do we learn? Paulo Júlio | José R. Maria
- 6|22 Survival of the fittest: Tourism Exposure and Firm Survival

Filipe B. Caires | Hugo Reis | Paulo M. M. Rodrigues

7|22 Mind the Build-up: Quantifying Tail Risks for Credit Growth in Portugal

> Ivan de Lorenzo Buratta | Marina Feliciano | Duarte Maia

- 8 22 Forgetting Approaches to Improve Forecasting Robert Hill | Paulo M. M. Rodrigues
- 9|22 Determinants of Cost of Equity for listed euro area banks Gabriel Zsurkis
- 10|22 Real effects of imperfect bank-firm matching Luísa Farinha | Sotirios Kokas | Enrico Sette | Serafeim Tsoukas

- 11|22 The solvency and funding cost nexus the role of market stigma for buffer usability Helena Carvalho | Lucas Avezum | Fátima Silva
- 12|22 Stayin' alive? Government support measures in Portugal during the Covid-19 pandemic Márcio Mateus | Katja Neugebauer
- 13|22
   Cross-Sectional Error Dependence in Panel

   Quantile Regressions
   Matei Demetrescu | Mehdi Hosseinkouchack |

   Paulo M. M. Rodrigues
- 14|22 Multinationals and services imports from havens: when policies stand in the way of tax planning Joana Garcia
- 15|22 Identification and Estimation of Continuous-Time Job Search Models with Preference Shocks Peter Arcidiacono | Attila Gyetvai | Arnaud Maurel | Ekaterina Jardim
- 16|22 Coworker Networks and the Role of Occupations in Job Finding Attila Gyetvai | Maria Zhu
- 17|22 What's Driving the Decline in Entrepreneurship? Nicholas Kozeniauskas
- 18|22 The Labor Share and the Monetary Transmission André Silva | João Gama | Bernardino Adão
- 19|22 Human capital spillovers and returns to education Pedro Portugal | Hugo Reis | Paulo Guimarães | Ana Rute Cardoso

20|22 Learning Through Repetition? A Dynamic Evaluation of Grade Retention in Portugal Emilio Borghesan | Hugo Reis | Petra E. Todd

## 2023

- 1|23 A single monetary policy for heterogeneous labour markets: the case of the euro area Sandra Gomes | Pascal Jacquinot | Matija Lozej
- 2|23 Price elasticity of demand and risk-bearing capacity in sovereign bond auctions Rui Albuquerque | José Miguel Cardoso-Costa | José Afonso Faias
- 3 23 A macroprudential look into the risk-return framework of banks' profitability Joana Passinhas | Ana Pereira
- 4 23 Mortgage Borrowing Caps: Leverage, Default, and Welfare João G. Oliveira | Leonor Queiró
- 5|23 Does Green Transition promote Green Innovation and Technological Acquisitions? Udichibarna Bose | Wildmer Daniel Gregori | Maria Martinez Cillero
- 6|23 Tail Index Estimation in the Presence of Covariates: Stock returns' tail risk dynamics João Nicolau | Paulo M.M. Rodrigues | Marian Z. Stoykov
- 7|23 The impact of ICT adoption on productivity: Evidence from Portuguese firm-level data João Amador | Cátia Silva
- 8|23 To use or not to use? Capital buffers and lending during a crisis Lucas Avezum | Vítor Oliveira | Diogo Serra
- 9|23 First passage times in portfolio optimization: a novel nonparametric approach Gabriel Zsurkis | João Nicolau | Paulo M. M. Rodrigues

- 21|22 Retrieving the Returns to Experience, Tenure, and Job Mobility from Work Histories John T. Addison | Pedro Portugal | Pedro Raposo
- 10|23Remote Work, Foreign Residents, and the<br/>Future of Global Cities<br/>João Guerreiro | Sérgio Rebelo | Pedro Teles
- 11|23 Trends and cycles during the COVID-19 pandemic period Paulo Júlio | José R. Maria
- 12|23 Gone with the wind: A structural decomposition of carbon emissions Fátima Cardoso | António Rua
- 13|23The Covid-19 Pandemic, Sovereign LoanGuarantees, and Financial Stability.Ivan De Lorenzo Buratta | Tiago Pinheiro
- 14|23 Prudential policy treatments to the COVID-19 economic crisis: an assessment of the effects. Ivan De Lorenzo Buratta | Diana Lima | Duarte Maia
- 15|23Structural and cyclical capital instruments in<br/>the 3D model: a simulation for Portugal<br/>Diana Lima | Duarte Maia | Ana Pereira
- 16|23 Currency choices and the role of the U.S. dollar in international services trade João Amador | Joana Garcia
- 17|23 Navigating with a compass: Charting the course of underlying inflation Nuno Lourenço | João Quelhas | António Rua
- 18|23 The determinants of the loss given default of residential mortgage loans in Portugal Márcio Mateus | Ângelo Ramos
- 19|23Regulating Artificial IntelligenceJoão Guerreiro | Sérgio Rebelo | Pedro Teles

- 20|23 The Impact of a Higher Cost of Credit on Exporters: Evidence from a Change in Banking Regulation João Monteiro | Pedro Dias Moreira
- 21|23 Taxes and Migration Flows: Preferential Tax Schemes for High-Skill Immigrants João Brogueira de Sousa | Pedro Teles

### 2024

1|24 Time-varying effects of monetary and macroprudential policies: does high inflation matter?

Wildmer Daniel Gregori | Ângelo Ramos

- 2 | 24 Behavioral Sticky Prices Sérgio Rebelo | Pedro Teles | Miguel Santana
- 3|24 The Monetary Financing of a Large Fiscal Shock Pedro Teles | Oreste Tristani
- 4|24 A Temporary VAT Cut in Three Acts: Announcement, Implementation, and Reversal

Tiago Bernardino | Ricardo Duque Gabriel | João Quelhas | Márcia Silva-Pereira

5|24 Risk and heterogeneity in benefits from vocational versus general secondary education: estimates for early and mature career stages

Joop Hartog | Pedro Raposo | Hugo Reis

6|24 Bayesian smoothing for time-varying extremal dependence

> Junho Lee | Miguel de Carvalho | António Rua | Julio Avila;

7|24 Parental Investments and Socio-Economic Gradients in Learning across European Countries

> Pedro Carneiro | Hugo Reis | Alessandro Toppeta;

22|23 What lies behind returns to schooling: the role of labor market sorting and worker heterogeneity Pedro Portugal | Hugo Reis | Paulo Guimarães Ana Rute Cardoso;

- 8|24 On the Asymmetrical Sensitivity of the Distribution of Real Wages to Business Cycle Fluctuations Rodrigo Barrela | Eduardo Costa | Pedro Portugal
- 9|24 Corporate Reorganization and the Reallocation of Labor in Bankruptcy Diana Bonfim | Gil Nogueira
- 10|24 The role of firms' characteristics on banks' interest rates Jaime Leyva
- 11 | 24 Health Shocks, Social Insurance, and Firms Anikó Bíró | István Boza | Attila Gyetvai | Dániel Prinz
- 12|24 A simple but powerful tail index regression Nicolau João | Paulo M. M. Rodrigues
- 13|24 Assessing the Macroeconomic Effects of IPCC Scenarios: Mitigation, Adaptation, and Carbon Sinks Bernardino Adão L António Antunes L Nuno

Bernardino Adão | António Antunes | Nuno Lourenço

14|24 Monetary policy and growth-at-risk: the role of institutional quality

Lorenz Emter | Afonso S. Moura | Ralph Setzer | Nico Zorell

15|24 Consumer Price-Setting Behaviour: Evidence from Food CPI Microdata

Fernando Martins | João Quelhas

16|24 Treasure Islands, Real Jobs? Workers and Anti-Avoidance Policies in a Tax Paradise

> Sónia Cabral | Joana Garcia | Raquel Miranda | Susana Peralta | João Pereira dos Santos

- 17|24 The welfare cost of an inflation shock on consumers: Who loses the most? Sónia Cabral | António Rua
- 18|24 Post-pandemic inflation dynamics in Portugal: an application of the Bernanke-Blanchard model Nuno Vilarinho Goncalves
- 19|24 On the band spectral estimation of business cycle models Nikolay Iskrev
- 20|24 Distance to Export: A Machine Learning Approach with Portuguese Firms Paulo Barbosa | João Amador | João Cortes
- 21|24 Export Promotion with Matchmaking and Grants: Evidence from Portuguese Firms Paulo Barbosa | João Amador | Esmeralda Arranhado